

Format for ANSWERING REVIEWERS



October 25, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: ESPS Manuscript NO: 13248).

Title: Consolidated and emerging inflammatory markers in coronary artery disease

Author: Valter Lubrano, Silvana Balzan

Name of Journal: *World Journal of Experimental Medicine*

ESPS Manuscript NO: 13248

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

(1) As suggested, we corrected the references

(2) A native speakers of English corrected the spelling and grammatical errors

(3) We corrected the sentences that you have reported on the base of the CrossCheck

Thank you again for publishing our manuscript in the *World Journal of Experimental Medicine*

Sincerely yours,

A handwritten signature in black ink, appearing to read 'Valter Lubrano'.

Dr. Valter Lubrano, PhD

Dept. of Laboratory Medicine

Fondazione G. Monasterio CNR-Regione Toscana,

56124 Pisa, Italy

Fax: +39-050-3153454

E-mail: walterl@ifc.cnr.it